Author’s response to reviews

Title: Offering Mental health first aid to a person with depression: A Delphi study to re-develop the guidelines published in 2008

Authors:
Kathy Bond (kathybond@mhfa.com.au)
Fairlie Cottrill (fairliec@mhfa.com.au)
Fiona Blee (fionab@mhfa.com.au)
Claire Kelly (clairek@mhfa.com.au)
Betty Kitchener (bettyakitchener@gmail.com)
Anthony Jorm (ajorm@unimelb.edu.au)

Version: 2 Date: 31 Oct 2018

Author’s response to reviews:

Dear editor,

Thank you for the opportunity to submit my article to your journal. I have run the last version of my article through a plagiarism checker and only 3 sentences came up as 'not unique' and they were all in the methods section, however, your email stated that you have picked up "an abundance of textual overlap with other previously published works. This overlap mainly exists in the Abstract, Methods, Results, Discussion, and Conclusions sections." I have changed the 3 sentences and the plagiarism checker tells me that all content is unique. If you are not happy with this version you will need to let me know what specific text you feel contain textual overlap so I can change them.

Kind Regards,

Kathy Bond